Author's response to reviews

Title: Factors affecting diagnosis and management of hypertension in Mazowe District of Mashonaland Central Province in Zimbabwe

Authors:

More Mungati (mungatim@yahoo.com)
Portia Manangazira (directoredc@gmail.com)
Lucia Takundwa (ltakundwa@gmail.com)
Notion T Gombe (gombent@yahoo.com)
Simbarashe Rusakaniko (srusakaniko@gmail.com)
Mufuta Tshimanga (mtshimanga@yahoo.com)

Version: 2 Date: 4 December 2013

Author's response to reviews: see over
Dear Team

RE: MANUSCRIPT FOR PEER REVIEW AND PUBLICATION

TITLE: Factors affecting diagnosis and management of hypertension in Mazowe District of Mashonaland Central Provinces in Zimbabwe

We have made the following changes made to the manuscript.

(1) Competing interests:

Please be advised that manuscripts must include a ‘Competing interests’ section. This should be placed after the Conclusions/Abbreviations. If there are none to declare, please include the statement “The authors declare that they have no competing interests.” Please consider the following questions and include an appropriate declaration of competing interests in your manuscript:

Response 1:

We added a section on competing interests after Conclusions and it reads “The authors declare that there are no competing interests”.

(2) We notice that you have included tables as additional files. If you want the tables to be visible within the final published manuscript please include them in the manuscript in a tables section following the references.
Response 2:

We added table 1 immediately after the references as advised

Yours faithfully

_____________________________

More Mungati, MBChB, MPH
E-mail: mungatim@yahoo.com